Massachusetts Institute of Technology Handout 23 6.857: Network and Computer Security October 29, 1998 Professor Ronald L. Rivest

Readings for Lecture 14

Attached are the following papers

• A Logic of Authentication [1].

• Kerberos: An Authentication Service for Computer Networks [2].

References

[1] Michael Burrows, Martin Abadi, and Roger Needham. A logic of authentication. ACM Transactions on Computer Systems, 8(1):18–36, February 1990.

[2] B. Clifford Neuman and Theodore Ts’o. Kerberos: An authentication service for computer networks. IEEE Communications Magazine, 32(9):33–38, September 1994.